

Geophysical Research Letters[•]

RESEARCH LETTER

10.1029/2024GL113786

Key Points:

- First evidence of peat initiation under late-20th century warming in a deglaciating montane valley in the Southern Hemisphere
- Consistent timing and direction of palaeoecological changes indicates catchment-scale ecological succession
- Climate-driven productivity increases may facilitate montane peatland development in suitable hydrological settings

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

R. E. Fewster, richardefewster.ac@gmail.com

Citation:

Fewster, R. E., Swindles, G. T., Carrivick, J. L., Gałka, M., Roland, T. P., McKeown, M., et al. (2025). Climate warming and deglaciation drive new peat formation in the Southern Alps, Aotearoa/New Zealand. *Geophysical Research Letters*, 52, e2024GL113786. https://doi.org/10.1029/ 2024GL113786

Received 20 NOV 2024 Accepted 27 JAN 2025

Author Contributions:

Conceptualization: R. E. Fewster, G. T. Swindles, J. L. Carrivick, M. McKeown, J. L. Sutherland, F. Tweed Formal analysis: R. E. Fewster, G. T. Swindles, J. L. Carrivick, M. Gałka, T. P. Roland, M. McKeown, J. L. Sutherland, F. Tweed, D. Mullan, C. Graham Funding acquisition: G. T. Swindles, J. L. Carrivick Writing - original draft: R. E. Fewster Writing - review & editing: R. E. Fewster, G. T. Swindles, J. L. Carrivick, M. Gałka, T. P. Roland, M. McKeown, J. L. Sutherland, F. Tweed, D. Mullan, C. Graham, A. Gallego-Sala, P. J. Morris

© 2025. The Author(s).

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Climate Warming and Deglaciation Drive New Peat Formation in the Southern Alps, Aotearoa/New Zealand

R. E. Fewster¹, G. T. Swindles^{1,2}, J. L. Carrivick³, M. Gałka⁴, T. P. Roland⁵, M. McKeown⁶, J. L. Sutherland⁷, F. Tweed⁸, D. Mullan¹, C. Graham¹, A. Gallego-Sala⁵, and P. J. Morris³

¹Geography and 14Chrono, School of Natural and Built Environment, Queen's University Belfast, Belfast, UK, ²Ottawa Carleton Geoscience Centre and Department of Earth Sciences, Carleton University, Ottawa, ON, Canada, ³School of Geography and water@Leeds, University of Leeds, Leeds, UK, ⁴Department of Biogeography, Paleoecology and Nature Conservation, Faculty of Biology and Environmental Protection, University of Lodz, Łodz, Poland, ⁵Geography, Faculty of Environment, Science and Economy, University of Exeter, Exeter, UK, ⁶Department of Geography, University College Cork, Cork, Ireland, ⁷School of Built Environment, Engineering and Computing, Leeds Beckett University, Leeds, UK, ⁸University of Staffordshire, Stoke-on-Trent, UK

Abstract Nascent peatlands represent an emerging, nature-based carbon sink in the global climate system. A warming climate and changing precipitation regime could drive peat initiation beyond the current latitudinal and altitudinal boundaries of the peatland bioclimatic envelope, through increases in plant productivity and moisture availability, with potential implications for global radiative forcing. However, contemporaneous observations of new peat formation remain scarce. We investigate peat initiation within the deglaciating Rob Roy valley in the Southern Alps, Aotearoa/New Zealand. We find that montane peats have developed across the head of the valley since ~1949 C.E., coinciding with regional climate warming and glacial retreat. Further, we identify a common ecological succession, characterized by a rise in brown mosses (mainly *Bryum*) beginning around ~1963 C.E. Our findings indicate the potential for wider peat expansion in increasingly warm and wet montane landscapes. However, further bioclimatic modeling is required to elucidate where future peatland developments may occur.

Plain Language Summary Peat, a highly organic soil, forms where plant matter is prevented from fully decomposing, forming an important carbon sink. 20th-century climate change appears to have made mountainous regions more favorable for new peat formation, because rising temperatures have enhanced plant growth and glacial retreat has exposed ice-free areas for ecological colonization. However, evidence of new peat forming in response to recent climate change is limited in the Southern hemisphere. We investigated peat formation in a deglaciating valley in the Southern Alps, Aotearoa/New Zealand. Our findings show that peat has been accumulating in the upper valley since ~1949 C.E., during a period of steady warming. From ~1963 C.E., plant communities across the valley have consistently shifted from herbaceous plants to mosses. Our results suggest that warming and wetting climates may enable future peat formation in mountainous regions, although further research is needed to determine the plausible extent of any future expansion.

1. Introduction

Global peatlands are an important carbon (C) store, containing >20% (~610 Gt) of the global soil organic carbon pool (Köchy et al., 2015; Page et al., 2011), despite covering only ~3% of the Earth's terrestrial surface (~4.2 million km²) (Xu et al., 2018). Peatlands have developed within a constrained range of climatic conditions, known as a bioclimate envelope (Morris et al., 2018). Peatlands form under a wide range of mean annual temperatures, but their distribution is also limited by precipitation and growing degree days (Fewster et al., 2020, 2022; Gallego-Sala & Colin Prentice, 2013; Yu et al., 2009). This bioclimate envelope, and thus global peatland distribution, has shifted throughout the Holocene (last ~11,750 years) due to deglaciation (Gorham et al., 2007) and postglacial climate change (Morris et al., 2018). Peat accumulation in high latitude and montane environments has been limited in part by low temperatures, which inhibit plant productivity. However, anthropogenic greenhouse gas release has increased global temperatures by ~1.1°C (1850–1900 to 2011–2020 C.E.; Lee et al., 2023), with faster warming occurring in mountainous and Arctic regions (e.g., Rantanen et al., 2022; Wang et al., 2014). Rising temperatures in montane regions have been associated with widespread greening (e.g., Carlson et al., 2017; Rumpf et al., 2022), shifts from snow to rainfall (e.g., Li et al., 2020; Serquet et al., 2011) and

changes in the timing and magnitude of snow and glacier melt (e.g., Klein et al., 2016; Vorkauf et al., 2021). These climate-driven changes may be favorable for new peat accumulation, because warmer, wetter climates stimulate plant growth, while glacial retreat exposes new environments for ecological colonization. Thus, the peatland bioclimate envelope may now be shifting into previously unsuitable landscapes. Any new peat formation may increase the global peatland C sink and partially offset decomposition losses from relict peatlands that now exist under unsuitable climates for peat growth (Gallego-Sala & Colin Prentice, 2013). However, observational evidence of recent peat formation is limited, particularly in the Southern Hemisphere, meaning that it is unclear how readily peatland extents might expand under warming climates.

The Southern Alps, Aotearoa/New Zealand, is a region undergoing rapid environmental changes in response to anthropogenic climate change. Glaciers presently occupy $\sim 1,021$ km² of the Southern Alps (Carrivick et al., 2020), but several glaciers have recently experienced rapid mass loss, including the Franz Josef and Fox glaciers (e.g., Anderson et al., 2006; Purdie et al., 2014), which have acted as barometers of climate change for the Southern Hemisphere. Since the Little Ice Age (LIA), ice volume loss across the Southern Alps has been up to six times greater than in other parts of the Southern mid-latitudes (Carrivick et al., 2020). This has resulted in rapid, proglacial landscape and habitat changes, including lake formation (Carrivick et al., 2022), alterations to river and fluvial sediment regimes (Carrivick & Tweed, 2021), and aquatic habitat conditioning (e.g., Fell et al., 2017). Recent climate change and continued deglaciation in the Southern Alps interior may have provided new, environmentally-plausible locations for peat initiation, which have not yet been accounted for in regional assessments and global modeling of peatland extents and soil organic carbon stocks. Montane climates in New Zealand are generally favorable for peat development, with high rainfall, low sunshine hours and low evaporation rates, although cool summers have often resulted in slow accumulation rates (McGlone, 2009). Typical montane peatlands in New Zealand include shallow sloping fens, bogs, and cushion bogs (McGlone, 2009). Peatlands are common in the neighboring, unglaciated district of Central Otago (Dickinson et al., 2002; McGlone et al., 1997) and peat has accumulated at the northwestern fringes of the Southern Alps since ~9,337 years B.P. (McGlone & Basher, 2012). However, peatland extents in the Southern Alps interior likely remain underestimated, because national assessments are often based on satellite mapping (e.g., Ausseil et al., 2008; Dymond et al., 2021) and many montane peatlands are small (<0.5 ha) and lack well-defined boundaries. New field-based studies are urgently needed to determine the location, timing and developmental pathways of emerging peatlands in this rapidly warming region. Establishing a case study of early peat development under anthropogenic climate change may provide useful insights into peatland dynamics in other montane and high latitude regions where similarly rapid bioclimatic changes are occurring.

In December 2018, we identified patches of peat-forming vegetation in a deglaciating valley in the Southern Alps of Aotearoa/New Zealand. Subsequent probing identified shallow underlying peat soils. In this study, we use a palaeoecological approach to investigate the timing of peat initiation and patterns of ecological succession in these poorly understood montane ecosystems.

2. Study Site and Methods

2.1. Study Site and Sampling

Our study site is located at 761 m above sea level (a.s.l.), near the upper lookout of the Rob Roy Track in the Mount Aspiring National Park (Table S1 in Supporting Information S1; Figure 1). This track terminates in front of Rob Roy hanging glacier, which is retreating slowly northwards toward Rob Roy Peak (2644 m a.s.l.) and thinning, now only being sustained by avalanching. Local vegetation, soil cover, and gullying suggest that the Rob Roy glacier once terminated at our study site during the LIA (~1450–1800 C.E.), while mapping of glacier outlines shows further retreat from 1978 C.E. to 2019 C.E. (Carrivick et al., 2020) (Figures 2a–2c).

The Rob Roy Stream drains the valley, flowing from a 500 m stretch of gravel outwash sediments at the toe of the glacier into a narrow step-pool channel that feeds the western Matukituki River. The altitudinal treeline (primarily native *Nothofagus spp.*) varies between 720 and 950 m a.s.l., due to hillslope mass movements near the Rob Roy Stream (Cadbury et al., 2011). Above the treeline, low-lying vascular plants grow atop scree, glacially-transported gravel, debris flow deposits and/or thin minerogenic soils (Radford et al., 2010). Previous studies have researched the hydroecology and thermodynamics of the Rob Roy Stream (Cadbury et al., 2008, 2011; Radford et al., 2010). Observations of geomorphological processes, deglaciation, vegetation regrowth and climate warming in this valley indicated potential for recent peat initiation. Our subsequent probing identified three





Figure 1. Maps showing: (a) our site location in the Southern Alps, Aotearoa/New Zealand; and (b) our sampling locations in the Rob Roy valley. Projection: New Zealand Transverse Mercator [NZTM2000]. Imagery from MAXAR and ESRI (2024).

disconnected peat patches, each located near a small watercourse, across a 15 m elevation gradient in the upper valley (Figure 1).

We sampled duplicate monoliths from the center of each peat patch (n = 6) to investigate local-scale heterogeneity in montane peat-forming processes (Table S1 in Supporting Information S1). Surface vegetation included mosses (e.g., *Bryum laevigatum*) and vascular plants (e.g., *Dracophyllum longifolium, Dolichoglottis lyallii, Celmisia gracilenta, Blechnum penna-marina* ssp. *alpina, Olearia nummulariifolia,* and *Gaultheria crassa*). We cut square-section monoliths to the peat base using a spade and serrated breadknife, dug a small trench immediately adjacent to the peat block, and sliced underneath using a pallet knife, allowing us to carefully lift the monolith. We packaged each peat monolith in clingfilm for transport to the University of Leeds, UK, where they were stored at ~4°C. We recorded pH at each sampling location using a calibrated field meter.



Geophysical Research Letters



Figure 2. Late-Holocene environmental change in our study region. Glacial outline data at: (a) regional, (b) landscape, and (c) local scales for the LIA (\sim 1450– \sim 1800 C. E.), 1978 C.E. and 2019 C.E. (adapted from Carrivick et al. (2020)). Panels (d–g) present historical (1836–2015 C.E.) and projected (2015–2100 C.E.) mean monthly temperature (°C) and precipitation (mm day⁻¹) trends for our study region (see methods for details).

2.2. Climate Data

To characterize historical climate change at our site, we downloaded monthly temperature and precipitation data for the period 1836–2015 C.E. from the NOAA/CIRES/DOE 20th Century Reanalysis V3 data set (Compo et al., 2011; Slivinski et al., 2019) from KNMI Climate Explorer (https://climexp.knmi.nl/). We analyzed data for the 1° latitude \times 1° longitude spatial domain overlying the study area (-43.5 to -44.5°N and 168.5 to 169.5°E).

To establish possible pathways of future climate change, we analyzed the Coupled Model Intercomparison Project 6 (CMIP6) mean for temperature and precipitation at the same study area for the period 1995–2100 C.E.. These data were downloaded from KNMI Climate Explorer for four shared socioeconomic pathway (SSP) scenarios: SSP1–2.6, SSP2–4.5, SSP3–7.0, and SSP5–4.5 (Riahi et al., 2017). Climate scenarios were bias corrected and spatially downscaled to the same spatial resolution of the reanalysis data set using the change factor method (e.g., Hawkins et al., 2013), by subtracting CMIP6 mean data for 1995–2014 C.E. from reanalysis data for the same period (1995–2014 C.E.), and then adding the difference (change factor) to CMIP6 mean data for 2015–2100 C.E. (on a monthly basis).

To assess the modern climatic conditions that facilitated peat initiation, we calculated mean annual temperature (MAT; °C) and total annual precipitation (APREC; mm yr⁻¹) values from the historical monthly means for the 30-year period surrounding peat initiation at our study site (1945–1974 C.E.; see Results below). We also calculated equivalent values for each SSP scenario for the period 2091–2100 C.E.. In Figure S1 in Supporting

Information S1, we compare the modern and projected climate space of our site to published data describing the palaeo-climate space of peatland initiation globally (n = 1,097 sites) (Morris et al., 2018).

2.3. Subsampling and Peat Properties

In the laboratory, we subsampled each peat monolith at contiguous 1 cm intervals using a stainless-steel knife, rinsed between cuts with deionized water to avoid contamination between samples. For each depth, we analyzed 1 cm^3 subsamples for dry bulk density (g cm⁻³), moisture content (%), and organic matter content (%) via loss-on-ignition, following Chambers et al. (2011).

2.4. Plant Macrofossils

We analyzed plant macrofossil compositions at 1 cm resolution for each monolith using a stereoscopic microscope, following Gałka et al. (2017). We recorded quantities of the primary peat-forming components (e.g., mosses, liverworts and vascular plants) as proportions of the total sample volume (%) and calculated absolute counts for fruits and seeds only. We present plant macrofossil proportions (%) in Figure 3, while full macrofossil diagrams (including counts of seeds and fruits) are presented in Figures S8–S13 in Supporting Information S1.

2.5. Chronology

We define basal peat to be any soils with >30% organic matter (e.g., Joosten and Clarke (2002)) to a depth of at least 0.1 m (Lourenco et al., 2022). While we acknowledge that this depth is shallower than some traditional peat definitions (Lourenco et al., 2022), we believe this is appropriate for our study of the earliest developmental phases of emerging peat patches, which have not had sufficient time to accumulate thick peat layers.

To establish robust chronologies for each peat monolith, we constructed Bayesian age-depth models from ²¹⁰Pb profiles and radiocarbon (¹⁴C) dating in Plum v.0.4.0 (Aquino-López et al., 2018) in R v.4.3.2 (R Core Team, 2022). To begin, we ¹⁴C dated the deepest available peat sample, then added additional ¹⁴C dates to further constrain our chronologies based on palaeoecological profiles and peat properties data. We primarily ¹⁴C dated plant macrofossils, and dated one bulk peat sample where suitable, intact plant material was unavailable (see Table S2 in Supporting Information S1 for details). All ¹⁴C dating was conducted using accelerated mass spectrometry (AMS) at the André E. Lalonde AMS Laboratory at the University of Ottawa. We calibrated all ¹⁴C dates using either the IntCal20 radiocarbon calibration curve (Reimer et al., 2020) or the SH1-2 post-bomb calibration curve (Hua et al., 2022). We also analyzed ²¹⁰Pb in continuous 1 cm³ peat subsamples spanning the full length of each monolith using alpha spectrometry at the University of Exeter, following Estop-Aragonés et al. (2018). We established calibrated ages for each peat sample from the probability estimates generated by our Bayesian age-depth models, and report weighted mean ages hereafter, unless otherwise specified. Prior settings and full modeled age estimates for each monolith are presented in Figures S2–S7 in Supporting Information S1.

3. Results and Discussion

Peat initiation and plant macrofossil accumulation began between ~1949 and ~1973 cal. yr C.E. across the deglaciating Rob Roy valley in the Southern Alps, Aotearoa/New Zealand. Regional climatic conditions at this time (MAT = 9.7° C; APREC = 1984 mm yr⁻¹; averaging period: 1945–1974 C.E.) fell well within the palaeo-climate envelope for global peatland initiation (Morris et al., 2018) (Figure S1 in Supporting Information S1). Our findings indicate a continuation of new peatland initiation in Aotearoa/New Zealand, which has been ongoing since ~14,000 cal. yr B.P. (McGlone, 2009). Similar evidence of 20th-century peat initiation has been observed in warming and wetting circum-Arctic regions, including subarctic Canada (Piilo et al., 2019) and Svalbard, Norway (Juselius et al., 2022). Peat moss banks in Antarctica have also evidenced increased productivity and vertical accumulation in response to late-20th century warming (Amesbury et al., 2017; Yu et al., 2016), although these Antarctic peats first formed as early as ~2,750 cal. yr BP (Loisel et al., 2017). To our knowledge, our study has recorded the most recent peat initiation dates for any montane peatlands in the Southern Hemisphere.

Modeled mean ages indicate that peat developed earliest at locations N2 and C2 (both ~1949 cal. yr C.E.), followed by C1 (~1957 cal. yr C.E.), N1 (~1958 cal. yr C.E.), S1 (~1959 cal. yr C.E.) and most recently at S2 (~1973 cal. yr C.E.) (Table S2 and Figures S2–S7 in Supporting Information S1). In basal peat samples, organic





Figure 3. Summarized palaeoecological information. Line graphs show dry bulk density (DBD; $g \text{ cm}^{-3}$) and organic matter content via loss on ignition (OM; %) (line graphs). Dashed line indicates interpreted depths of basal peat (see methods for details). Stacked bar charts show percentage composition of the major peat-forming components. For detailed palaeoecological diagrams see Figures S8–S13 in Supporting Information S1.

matter content was highest in the N1 and N2 monoliths (67% and 69%), while the basal samples of C1 and S1 contained a lower organic matter content that was closer to our threshold for peat soils (35% and 34%, respectively). Our interpretations of the peat-mineral boundary were supported by low dry bulk density values in all basal samples (0.07–0.18 g cm⁻³), which is well below the typical upper limits for peat soils, for example, 0.3 g cm⁻³ (Morris et al., 2022). Indeed, deeper, non-peat samples in monoliths N2, C1, C2, and S2 substantially exceeded this dry bulk density threshold (Figure 3), while monoliths N1 and S1 were inferred to be entirely composed of peat, having formed directly on a matrix of superficial glaci-fluvial silts and sands, moraine gravel and boulders, hillslope scree and debris flow deposits. Our age-depth models indicated varying rates of recent peat and C accumulation, with thicker peat deposits at N1 (25 cm), N2 and C1 (both 22 cm) and shallower peats down-valley at C2, S1 and S2 (14, 11 and 12 cm, respectively). However, we do not interpret patterns of peat accumulation further, because of the difficulties in untangling genuine temporal variability in individual records from incomplete and secondary decomposition (see Young et al., 2021).

Our plant macrofossil analyses suggest a consistency in ecological succession across our sampling locations toward Bryum mosses and/or liverworts (Figure 3; Figures S8–S13 in Supporting Information S1). In five of the six monoliths (N2, C1, C2, S1 and S2), herbaceous leaves and roots comprised the primary peat-forming vegetation in early peat layers (55%-100% composition). Moderate proportions of brown mosses developed for a temporary period after peat formation in monoliths N1 and S2. Conversely, in the northernmost site, N1, early peats are dominated by liverworts (\geq 80%), with only minor proportions (\leq 10%) of herbaceous material recorded in this profile until present, when abundances increased to 60%. Because of the macrofossil composition of these samples, we infer that a wet, nutrient-rich, minerotrophic environment supported the early stages of peat formation. In all monoliths, counts of Apiaceae seeds increase during, or shortly after, the primary palaeoecological transitions. In monoliths N1 and N2, Bryum cf. leavigatum became dominant from ~1967 cal. yr C.E. and ~1993 cal. yr C.E., respectively, with stable proportions of 60%–90% until present. Equivalent, recent increases in Bryum mosses occurred in S1 and S2 after ~2011 cal. yr C.E., alongside increases in organic matter content and reductions to dry bulk density. By contrast, Bryum mosses were broadly absent from the C1 and C2 records, which instead indicated increased abundances of liverworts (C1 and C2) and brown mosses (C2 only) after ~2012 cal. yr C.E., a pattern which is not reflected in our other monoliths. Bryum mosses and liverworts are resilient, pioneer plants known to thrive on deglaciated substrates where competition from vascular plants is limited (Wietrzyk-Pełka et al., 2020), and indicate that conditions are still minerotrophic (Van der Putten et al., 2008, 2012), an assertion supported by in-field pH measurements of 6.38–6.70 at our sampling locations (Table S1 in Supporting Information S1). Similar increases in *Bryum* and liverwort abundance have previously been recorded during the early developmental stages of some older subantarctic peatlands (e.g., Van der Putten et al., 2008, 2012). However, decay rates of liverworts vary widely (Lang et al., 2009) and some liverwort remains may eventually fully decompose and not be incorporated into deeper, older peats (Svensson, 1986). Therefore, in the long term, peat layers formed from moss, herbaceous and woody macrofossils may preferentially persist (Figure 3), because these plant litter are generally more resilient to decomposition.

The absence of *Sphagnum* macrofossils in our monoliths is likely due to the distinct ecohydrological conditions of New Zealand peatlands, which differ substantially from Northern Hemisphere bogs. Unlike their Northern counterparts, montane cushion bogs of Aotearoa/New Zealand do not typically support *Sphagnum* during their natural succession. More commonly, cushion bogs are dominated by graminoids, herbs, subshrubs and restiads (McGlone, 2009; McGlone et al., 1997). McGlone and Wilmshurst (1999) have suggested that *Sphagnum* dominance in some lower-lying, South Island bogs may be a recent phenomenon, driven by anthropogenic influences. For example, deforestation through Māori burning practices and European logging has likely altered the hydrology of many catchments, increasing water flow to lower-elevation peatlands and creating wetter conditions favorable to *Sphagnum* proliferation (McGlone, 2009). Recent research by McKeown et al. (2024) also suggests that *Sphagnum* dominance in some Aotearoa/New Zealand peatlands may not reflect natural conditions but rather anthropogenic alteration. This highlights the unique dynamics of Aotearoa/New Zealand's peatland ecosystems and underscores the need for region-specific approaches to ecological assessments, avoiding the assumption that these emerging peatlands will follow Northern Hemisphere models of ecological succession.

Our observations of 20th-century peat initiation and early ecological succession in the Rob Roy valley coincides with a period of almost linear warming, with MAT increasing by >1.5°C during the post-industrial period and by 0.46°C during the 30-year period from 1986-1995 C.E. to 2006-2015 C.E. (Figure 2d). Concurrent precipitation increases have improved the climatic suitability of our study area for new peat initiation (Figure 2f). Further increases in temperature and precipitation have been projected for this region by 2100 C.E., even under a "best-case" climate-mitigation scenario (SSP1-2.6) (Figures 2e and 2g). Under a "worst-case" no-mitigation scenario (SSP5-8.5), climatic conditions at our site are projected to shift toward the limits of the palaeo-climate space for global peatland initiation, but would still remain suitable by 2091-2100 C.E. (Morris et al., 2018) (Figure S1 in Supporting Information S1). Although recent, temperature-driven increases in plant productivity have driven advancement of altitudinal treelines in alpine regions worldwide (e.g., Harsch et al., 2009), treelines in the Southern Alps have instead exhibited remarkable stability (e.g., Cullen et al., 2001; Hansson et al., 2023). Nonetheless, late-Holocene climate change and subsequent deglaciation seem to have stimulated recent peat initiation in this valley. Warming growing seasons exert a primary control on peat initiation (Morris et al., 2018), while ice retreat in this valley has exposed a pristine, unoccupied terrain for ecological colonization and subsequent peat growth. Similar environmental changes appear to have accelerated the growth of emerging peat patches in Arctic Alaska (Cleary et al., 2024; Gałka

et al., 2018, 2023; Taylor et al., 2019), while satellite observations indicate that sustained greening and moss expansion is ongoing across warming regions of the Arctic (e.g., Crichton et al., 2022; Grimes et al., 2024; Karlsen et al., 2024) and Antarctic (Roland et al., 2024). Holocene records indicate that peat development in deglaciated landscapes of North America and northern Europe has commonly lagged ice retreat by several millennia (Gorham et al., 2007; Ruppel et al., 2013). However, evidence of LIA moraines within meters of our sampling locations (Figures 2a–2c; Carrivick et al., 2020), coupled with the late 20th-century timing of our basal peat samples (Figure 1; Table S2 in Supporting Information S1), suggest that any postglacial lag in peat development in the Rob Roy valley has been by no more than a few centuries.

In addition to sufficient warmth for plant productivity, sustained peat accumulation requires a landscape setting with a persistent excess of surface moisture, resulting from excess precipitation and/or a slowly draining substrate, which inhibits the complete decomposition of plant litter (e.g., Clymo, 1984). Basal peats at our sites were primarily composed of herbaceous taxa that thrive in moist or wet minerotrophic conditions (Figure 3). Our studied peat patches also formed in similar topographical locations, particularly undisturbed, west-facing, gentle slopes of no more than 15° with small streams that receive a regular supply of meltwater, either from the northward Rob Roy glacier or adjacent snowpacks on the eastern peaks (Figure 1). Peat was absent in the intervening and upslope areas from our sampling locations, as well as the steep cliffs along the western valley side and along the Rob Roy stream where intense geomorphological action is ongoing, resulting in bank erosion, instability and subsidence. Topographical and hydrological constraints on peat growth may also explain why peat patches do not currently extend as far upslope as the present altitudinal treeline (Figure 1). It is possible that the underlying geological matrix of silt, gravel and rocks across much of this valley is too porous to retain sufficient moisture to support peat growth from precipitation alone, without additional inputs from snow and glacier melt. The observed concentration of these shallow peats around meltwater streams places their long-term resilience in doubt if the Rob Roy glacier continues to retreat and ultimately disappears. However, projected future increases in precipitation (Figure 2d), alongside shifts from snow to rainfall, may partially overcome this hydrological constraint on peat development by increasing surface wetness in certain parts of the valley. Further bioclimatic modeling, supported by highresolution geological and topographical data sets, is now recommended to elucidate how projected climate change may impact the extent of potential peat-forming areas in this montane environment.

The newly formed montane peats we have identified in Aotearoa/New Zealand represent a small, albeit growing, natural C sink that has not yet been accounted for in large-scale peatland maps (e.g., Xu et al., 2018) or landsurface model projections (e.g., Hardouin et al., 2024; Müller & Joos, 2021). It is therefore possible that coarse-scale model projections presently underestimate the capacity for warming and deglaciating montane regions to sustain new peat growth and thus enhanced C sequestration. Continued expansion of these, and other, montane peat patches could partially offset greenhouse gas losses from the extensive anthropogenic degradation of lowland peatlands, which have declined in areal extent by ~90% in Aotearoa/New Zealand since European settlement (Ausseil et al., 2008; McGlone, 2009). Indeed, more-developed montane peatlands in Aotearoa/New Zealand have been found to reach substantial thicknesses (>2 m), even if their areal extent remains small (e.g., McGlone, 2009; McGlone et al., 1995).

The easy accessibility of our site, located at the northern end of the Rob Roy Glacier track, makes it a promising case study for real-time monitoring of the response of nascent peats to glacial retreat and climate warming. Additionally, future studies could determine rates of lateral peat expansion at this site using well-dated, corebased transects from the peat margins (e.g., Juselius-Rajamäki et al., 2023). Further surveying is also now needed to elucidate the wider spatial context of our small-scale case study, which has considered six peat monoliths in a single deglaciating valley. Remote sensing techniques could scale-up the identification of plausible areas for recent peat initiation by assessing changes in surface moisture and vegetation productivity (e.g., Crichton et al., 2022). To aid this effort, we propose a criteria-based framework for identifying potentially suitable locations for montane peat initiation, based on our palaeoecological and field-based observations. Our findings suggest that peat can initiate in montane environments that exhibit: (a) areas susceptible to ecological colonization, such as bare mineral substrates or postglacial melt ponds (as found in deglaciating valleys); (b) suitable climatic conditions for sustained plant growth, including temperature and precipitation; and (c) saturated surface conditions, arising from some combination of gently sloping terrain, poorly drained substrate, presence of freshwater streams and/or meltwater. We recommend that future studies employ this framework to identify possible locations for new peat initiation in other, similar montane regions. As shown in our study, combining



such predictions with ground-truthed observations and palaeoecological analyses of early ecological succession can provide new insights for constraining model representations of the dynamics of peatland development under a changing climate.

4. Conclusions

Our study presents, to our knowledge, the first evidence of new montane peat initiation occurring under 20thcentury climate change in the Southern Hemisphere. Our radiometric dating of basal peat samples indicates that peat initiation in the deglaciating Rob Roy valley in the Southern Alps, Aotearoa/New Zealand began at ~1949 cal. yr C.E. and has continued until at least ~1973 cal. yr C.E.. Our palaeoecological analysis indicates that these nascent peats have undergone a consistent trajectory of ecological succession, with increasing abundances of *Bryum* moss evident in five of the six monoliths. Continued expansion of these emerging peat patches may increase the C-sink capacity of global peatlands, thereby enhancing their overall climate mitigation effect. However, the apparent dependence of these shallow peats on meltwater availability, coupled with ambiguity regarding the extent of suitable niches for broader montane peat expansion, makes the wider spatial implications of our findings uncertain. Further observational and palaeoecological research is therefore needed to identify similar examples of recent and ongoing peat formation in other montane regions to improve understanding of peat initiation dynamics at the altitudinal boundaries of their bioclimatic envelope. The full geographic extent of new peat formation driven by recent climate change remains unknown, as does any potential impact on the global C cycle.

Data Availability Statement

The palaeoecological data underlying this research are available at Zenodo via https://zenodo.org/records/ 14177780 (Fewster, 2025).

References

- Amesbury, M. J., Roland, T. P., Royles, J., Hodgson, D. A., Convey, P., Griffiths, H., & Charman, D. J. (2017). Widespread biological response to rapid warming on the Antarctic Peninsula. *Current Biology*, 27(11), 1616–1622.e2. https://doi.org/10.1016/j.cub.2017.04.034
- Anderson, B., Lawson, W., Owens, I., & Goodsell, B. (2006). Past and future mass balance of "Ka Roimata o Hine Hukatere" Franz Josef Glacier, New Zealand. Journal of Glaciology, 52(179), 597–607. https://doi.org/10.3189/172756506781828449
- Aquino-López, M. A., Blaauw, M., Christen, J. A., & Sanderson, N. K. (2018). Bayesian analysis of 210Pb dating. Journal of Agricultural, Biological, and Environmental Statistics, 23(3), 317–333. https://doi.org/10.1007/s13253-018-0328-7
- Ausseil, A.-G., Gerbeaux, P., Chadderton, W. L., Stephens, T., Brown, D., & Leathwick, J. (2008). Wetland ecosystems of national importance for biodiversity: Criteria, methods and candidate list of nationally important inland wetlands. *Landcare Research Contract Report LC0708158*.
- Cadbury, S. L., Hannah, D. M., Milner, A. M., Pearson, C. P., & Brown, L. E. (2008). Stream temperature dynamics within a New Zealand Glacierized River Basin. *River Research and Applications*, 24(1), 68–89. https://doi.org/10.1002/rra.1048
- Cadbury, S. L., Milner, A. M., & Hannah, D. M. (2011). Hydroecology of a New Zealand glacier-fed river: Linking longitudinal zonation of physical habitat and macroinvertebrate communities. *Ecohydrology*, 4, 520–531. https://doi.org/10.1002/eco.185
- Carlson, B. Z., Corona, M. C., Dentant, C., Bonet, R., Thuiller, W., & Choler, P. (2017). Observed long-term greening of alpine vegetation—A case study in the French Alps. *Environmental Research Letters*, 12(11), 114006. https://doi.org/10.1088/1748-9326/aa84bd
- Carrivick, J. L., James, W. H. M., Grimes, M., Sutherland, J. L., & Lorrey, A. M. (2020). Ice thickness and volume changes across the Southern Alps, New Zealand, from the little ice age to present. *Scientific Reports*, 10(1), 13392. https://doi.org/10.1038/s41598-020-70276-8
- Carrivick, J. L., Sutherland, J. L., Huss, M., Purdie, H., Stringer, C. D., Grimes, M., et al. (2022). Coincident evolution of glaciers and ice-marginal proglacial lakes across the Southern Alps, New Zealand: Past, present and future. *Global and Planetary Change*, 211, 103792. https://doi.org/ 10.1016/j.gloplacha.2022.103792
- Carrivick, J. L., & Tweed, F. S. (2021). Deglaciation controls on sediment yield: Towards capturing spatio-temporal variability. *Earth-Science Reviews*, 221, 103809. https://doi.org/10.1016/j.earscirev.2021.103809
- Chambers, F. M., Beilman, D. W., & Yu, Z. (2011). Methods for determining peat humification and for quantifying peat bulk density, organic matter and carbon content for palaeostudies of climate and peatland carbon dynamics 10.
- Cleary, K. G., Xia, Z., & Yu, Z. (2024). The growth and carbon sink of tundra peat patches in Arctic Alaska. *Journal of Geophysical Research: Biogeosciences*, 129(6), e2023JG007890. https://doi.org/10.1029/2023JG007890
- Clymo, R. (1984). The limits to peat bog growth. *Philosophical Transactions of the Royal Society of London B Biological Sciences*, 303, 605–654.
 Compo, G. P., Whitaker, J. S., Sardeshmukh, P. D., Matsui, N., Allan, R. J., Yin, X., et al. (2011). The twentieth century reanalysis project. *Quarterly Journal of the Royal Meteorological Society*, 137(654), 1–28. https://doi.org/10.1002/qj.776
- Crichton, K. A., Anderson, K., Charman, D. J., & Gallego-Sala, A. (2022). Seasonal climate drivers of peak NDVI in a series of Arctic peatlands. Science of the Total Environment, 838, 156419. https://doi.org/10.1016/j.scitotenv.2022.156419
- Cullen, L. E., Stewart, G. H., Duncan, R. P., & Palmer, J. G. (2001). Disturbance and climate warming influences on New Zealand Nothofagus tree-line population dynamics. *Journal of Ecology*, 89(6), 1061–1071. https://doi.org/10.1111/j.1365-2745.2001.00628.x
- Dickinson, K. J., Chagué-Goff, C., Mark, A. F., & Cullen, L. (2002). Ecological processes and trophic status of two low-alpine patterned mires, south-central South Island, New Zealand. Austral Ecology, 27(4), 369–384. https://doi.org/10.1046/j.1442-9993.2002.01191.x
- Dymond, J. R., Sabetizade, M., Newsome, P. F., Harmsworth, G. R., & Ausseil, A.-G. (2021). Revised extent of wetlands in New Zealand. New Zealand Journal of Ecology, 45, 1–8. https://doi.org/10.20417/nzjecol.45.32

Richard Fewster and Graeme Swindles (PI) acknowledge the support of a Leverhulme Trust Research Project Grant (RPG-2021-354). Jonathan Carrivick and Graeme Swindles received funding from the School of Geography, University of Leeds for logistics and teaching of the New Zealand Field Trip, Jenna Sutherland was funded by a PhD studentship from the Natural Environment Research Council (NERC) SPHERES Doctoral Training Partnership (NE/L002574/1), Jenna Sutherland also acknowledges funding from the water@Leeds SPRING Award. Fiona Tweed acknowledges funding from the AdvanceHE National Teaching Fellowship Scheme, which enabled her participation in New Zealand fieldwork. David Wilson conducted peat properties analysis in University of Leeds laboratories.

- Estop-Aragonés, C., Czimczik, C. I., Heffernan, L., Gibson, C., Walker, J. C., Xu, X., & Olefeldt, D. (2018). Respiration of aged soil carbon during fall in permafrost peatlands enhanced by active layer deepening following wildfire but limited following thermokarst. *Environmental Research Letters*, 13(8), 085002. https://doi.org/10.1088/1748-9326/aad5f0
- Fell, S. C., Carrivick, J. L., & Brown, L. E. (2017). The multitrophic effects of climate change and glacier retreat in Mountain Rivers. *BioScience*, 67(10), 897–911. https://doi.org/10.1093/biosci/bix107
- Fewster, R. E. (2025). Supporting data for: Climate warming and deglaciation drive new peat formation in the Southern Alps, Aotearoa/New Zealand (Fewster et al.). https://doi.org/10.5281/zenodo.14177780
- Fewster, R. E., Morris, P. J., Ivanovic, R. F., Swindles, G. T., Peregon, A. M., & Smith, C. J. (2022). Imminent loss of climate space for permafrost peatlands in Europe and Western Siberia. *Nature Climate Change*, 12(4), 373–379. https://doi.org/10.1038/s41558-022-01296-7
- Fewster, R. E., Morris, P. J., Swindles, G. T., Gregoire, L. J., Ivanovic, R. F., Valdes, P. J., & Mullan, D. (2020). Drivers of Holocene palsa distribution in North America. *Quaternary Science Reviews*, 240, 106337. https://doi.org/10.1016/j.quascirev.2020.106337
- Gałka, M., Diaconu, A.-C., Cwanek, A., Hedenäs, L., Knorr, K.-H., Kołaczek, P., et al. (2023). Climate-induced hydrological fluctuations shape Arctic Alaskan peatland plant communities. Science of the Total Environment, 905, 167381. https://doi.org/10.1016/j.scitotenv.2023.167381
- Gałka, M., Swindles, G. T., Szal, M., Fulweber, R., & Feurdean, A. (2018). Response of plant communities to climate change during the late Holocene: Palaeoecological insights from peatlands in the Alaskan Arctic. *Ecological Indicators*, 85, 525–536. https://doi.org/10.1016/j. ecolind.2017.10.062
- Gałka, M., Tobolski, K., Lamentowicz, Ł., Ersek, V., Jassey, V. E. J., van der Knaap, W. O., & Lamentowicz, M. (2017). Unveiling exceptional Baltic bog ecohydrology, autogenic succession and climate change during the last 2000 years in CE Europe using replicate cores, multi-proxy data and functional traits of testate amoebae. *Quaternary Science Reviews*, 156, 90–106. https://doi.org/10.1016/j.quascirev.2016.11.034
- Gallego-Sala, A. V., & Colin Prentice, I. (2013). Blanket peat biome endangered by climate change. Nature Climate Change, 3(2), 152–155. https://doi.org/10.1038/nclimate1672
- Gorham, E., Lehman, C., Dyke, A., Janssens, J., & Dyke, L. (2007). Temporal and spatial aspects of peatland initiation following deglaciation in North America. *Quaternary Science Reviews*, 26(3–4), 300–311. https://doi.org/10.1016/j.quascirev.2006.08.008
- Grimes, M., Carrivick, J. L., Smith, M. W., & Comber, A. J. (2024). Land cover changes across Greenland dominated by a doubling of vegetation in three decades. *Scientific Reports*, 14(1), 3120. https://doi.org/10.1038/s41598-024-52124-1
- Hansson, A., Shulmeister, J., Dargusch, P., & Hill, G. (2023). A review of factors controlling Southern Hemisphere treelines and the implications of climate change on future treeline dynamics. Agricultural and Forest Meteorology, 332, 109375. https://doi.org/10.1016/j.agrformet.2023. 109375
- Hardouin, L., Decharme, B., Colin, J., & Delire, C. (2024). Climate-driven projections of future global wetlands extent. *Earth's Future*, 12(9), e2024EF004553. https://doi.org/10.1029/2024EF004553
- Harsch, M. A., Hulme, P. E., McGlone, M. S., & Duncan, R. P. (2009). Are treelines advancing? A global meta-analysis of treeline response to climate warming. *Ecology Letters*, 12(10), 1040–1049. https://doi.org/10.1111/j.1461-0248.2009.01355.x
- Hawkins, E., Osborne, T. M., Ho, C. K., & Challinor, A. J. (2013). Calibration and bias correction of climate projections for crop modelling: An idealised case study over Europe. Agricultural and Forest Meteorolog, 170, 19–31. https://doi.org/10.1016/j.agrformet.2012.04.007
- Hua, Q., Turnbull, J. C., Santos, G. M., Rakowski, A. Z., Ancapichún, S., De Pol-Holz, R., et al. (2022). Atmospheric radiocarbon for the period 1950–2019. Radiocarbon, 64(4), 723–745. https://doi.org/10.1017/rdc.2021.95

Joosten, H., & Clarke, D. (2002). Wise use of mires and peatlands. International Mire Conservation Group / International Peat Society, 304.

- Juselius, T., Ravolainen, V., Zhang, H., Piilo, S., Müller, M., Gallego-Sala, A., & Väliranta, M. (2022). Newly initiated carbon stock, organic soil accumulation patterns and main driving factors in the High Arctic Svalbard, Norway. *Scientific Reports*, 12(1), 4679. https://doi.org/10.1038/ s41598-022-08652-9
- Juselius-Rajamäki, T., Väliranta, M., & Korhola, A. (2023). The ongoing lateral expansion of peatlands in Finland. Global Change Biology, 29(24), 7173–7191. https://doi.org/10.1111/gcb.16988
- Karlsen, S. R., Elvebakk, A., Stendardi, L., Høgda, K. A., & Macias-Fauria, M. (2024). Greening of Svalbard. Science of the Total Environment, 945, 174130. https://doi.org/10.1016/j.scitotenv.2024.174130
- Klein, G., Vitasse, Y., Rixen, C., Marty, C., & Rebetez, M. (2016). Shorter snow cover duration since 1970 in the Swiss Alps due to earlier snowmelt more than to later snow onset. *Climate Change*, 139(3–4), 637–649. https://doi.org/10.1007/s10584-016-1806-y
- Köchy, M., Hiederer, R., & Freibauer, A. (2015). Global distribution of soil organic carbon Part 1: Masses and frequency distributions of SOC stocks for the tropics, permafrost regions, wetlands, and the world. SOIL, 1, 351–365. https://doi.org/10.5194/soil-1-351-2015
- Lang, S. I., Cornelissen, J. H. C., Klahn, T., Van Logtestijn, R. S. P., Broekman, R., Schweikert, W., & Aerts, R. (2009). An experimental comparison of chemical traits and litter decomposition rates in a diverse range of subarctic bryophyte, lichen and vascular plant species. *Journal* of Ecology, 97(5), 886–900. https://doi.org/10.1111/j.1365-2745.2009.01538.x
- Lee, H., Calvin, K., Dasgupta, D., Krinner, G., Mukherji, A., Thorne, P., et al. (2023). IPCC, 2023: Climate change 2023: Synthesis report, summary for policymakers. In H. Lee & J. Romero (Eds.), Contribution of Working Groups I, II and III to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change [Core Writing Team. IPCC.
- Li, Z., Chen, Y., Li, Y., & Wang, Y. (2020). Declining snowfall fraction in the alpine regions, Central Asia. Scientific Reports, 10(1), 3476. https:// doi.org/10.1038/s41598-020-60303-z
- Loisel, J., Yu, Z., Beilman, D. W., Kaiser, K., & Parnikoza, I. (2017). Peatland ecosystem processes in the maritime Antarctic during warm climates. Scientific Reports, 7(1), 12344. https://doi.org/10.1038/s41598-017-12479-0
- Lourenco, M., Fitchett, J. M., & Woodborne, S. (2022). Peat definitions: A critical review. Progress in Physical Geography: Earth and Environment, 47(4), 506–520. https://doi.org/10.1177/03091333221118353
- MAXAR and ESRI. (2024). World imagery.
- McGlone, M. S., Moar, N., & Meurk, C. (1997). Growth and vegetation history of alpine mires on the old man range, central Otago, New Zealand. *Arctic and Alpine Research*, 29(1), 32–44. https://doi.org/10.2307/1551834
- McGlone, M. S. (2009). Postglacial history of New Zealand wetlands and implications for their conservation. *New Zealand Journal of Ecology*, 33, 1–23.
- McGlone, M. S., & Basher, L. (2012). Holocene vegetation change at treeline, Cropp Valley, Southern Alps, New Zealand. *Terra Australis*, 34, 343–358.
- McGlone, M. S., Mark, A. F., & Bell, D. (1995). Late Pleistocene and Holocene vegetation history, Central Otago, South Island, New Zealand. Journal of the Royal Society of New Zealand, 25, 1–22. https://doi.org/10.1080/03014223.1995.9517480
- McGlone, M. S., & Wilmshurst, J. M. (1999). A Holocene record of climate, vegetation change and peat bog development, east Otago, South Island, New Zealand. *Journal of Quaternary Science*, *14*(3), 239–254. https://doi.org/10.1002/(SICI)1099-1417(199905)14:3<239::AID-JQS438>3.0.CO;2-9

- McKeown, M. M., Burge, O. R., Richardson, S. J., Wood, J. R., Mitchell, E. A. D., & Wilmshurst, J. M. (2024). Biomonitoring tool for New Zealand peatlands: Testate amoebae and vascular plants as promising bioindicators. *Journal of Environmental Management*, 354, 120243. https://doi.org/10.1016/j.jenvman.2024.120243
- Morris, P. J., Davies, M., Baird, A., Balliston, N., Bourgault, M., Clymo, R., et al. (2022). Saturated hydraulic conductivity in northern peats inferred from other measurements. *Water Resources Research*, 58(11), e2022WR033181. https://doi.org/10.1029/2022wr033181
- Morris, P. J., Swindles, G. T., Valdes, P. J., Ivanovic, R. F., Gregoire, L. J., Smith, M. W., et al. (2018). Global peatland initiation driven by regionally asynchronous warming. *Proceedings of the National Academy of Sciences of the United States of America*, 115(19), 4851–4856. https://doi.org/10.1073/pnas.1717838115
- Müller, J., & Joos, F. (2021). Committed and projected future changes in global peatlands—Continued transient model simulations since the Last Glacial Maximum. *Biogeosciences*, *18*(12), 3657–3687. https://doi.org/10.5194/bg-18-3657-2021
- Page, S. E., Rieley, J. O., & Banks, C. J. (2011). Global and regional importance of the tropical peatland carbon pool. *Global Change Biology*, 17(2), 798–818. https://doi.org/10.1111/j.1365-2486.2010.02279.x
- Piilo, S. R., Zhang, H., Garneau, M., Gallego-Sala, A., Amesbury, M. J., & Väliranta, M. M. (2019). Recent peat and carbon accumulation following the Little Ice Age in northwestern Québec, Canada. *Environmental Research Letters*, 14(7), 075002. https://doi.org/10.1088/1748-9326/ab11ec
- Purdie, H., Anderson, B., Chinn, T., Owens, I., Mackintosh, A., & Lawson, W. (2014). Franz Josef and Fox Glaciers, New Zealand: Historic length records. *Global and Planetary Change*, 121, 41–52. https://doi.org/10.1016/j.gloplacha.2014.06.008
- Radford, I. J., Dickinson, K. J. M., & Lord, J. M. (2010). Does disturbance, competition or resource limitation underlie Hieracium lepidulum invasion in New Zealand? Mechanisms of establishment and persistence, and functional differentiation among invasive and native species. *Austral Ecology*, 35(3), 282–293. https://doi.org/10.1111/j.1442-9993.2009.02034.x
- Rantanen, M., Karpechko, A. Y., Lipponen, A., Nordling, K., Hyvärinen, O., Ruosteenoja, K., et al. (2022). The Arctic has warmed nearly four times faster than the globe since 1979. Communications Earth & Environment, 3, 1–10. https://doi.org/10.1038/s43247-022-00498-3
- R Core Team. (2022). R: The R project for statistical computing. Retrieved from https://www.r-project.org/
- Reimer, P. J., Austin, W. E. N., Bard, E., Bayliss, A., Blackwell, P. G., Ramsey, C. B., et al. (2020). The IntCal20 northern hemisphere radiocarbon age calibration curve (0–55 cal kBP). *Radiocarbon*, 62(4), 725–757. https://doi.org/10.1017/RDC.2020.41
- Riahi, K., van Vuuren, D. P., Kriegler, E., Edmonds, J., O'Neill, B. C., Fujimori, S., et al. (2017). The Shared Socioeconomic Pathways and their energy, land use, and greenhouse gas emissions implications: An overview. *Global Environmental Change*, 42, 153–168. https://doi.org/10. 1016/j.gloenvcha.2016.05.009
- Roland, T. P., Bartlett, O. T., Charman, D. J., Anderson, K., Hodgson, D. A., Amesbury, M. J., et al. (2024). Sustained greening of the Antarctic Peninsula observed from satellites. *Nature Geoscience*, 17(11), 1–6. https://doi.org/10.1038/s41561-024-01564-5
- Rumpf, S. B., Gravey, M., Brönnimann, O., Luoto, M., Cianfrani, C., Mariethoz, G., & Guisan, A. (2022). From white to green: Snow cover loss and increased vegetation productivity in the European Alps. Science, 376(6597), 1119–1122. https://doi.org/10.1126/science.abn6697
- Ruppel, M., Väliranta, M., Virtanen, T., & Korhola, A. (2013). Postglacial spatiotemporal peatland initiation and lateral expansion dynamics in North America and northern Europe. *The Holocene*, 23(11), 1596–1606. https://doi.org/10.1177/0959683613499053
- Serquet, G., Marty, C., Dulex, J.-P., & Rebetez, M. (2011). Seasonal trends and temperature dependence of the snowfall/precipitation-day ratio in Switzerland. *Geophysical Research Letters*, 38(7). https://doi.org/10.1029/2011GL046976
- Slivinski, L. C., Compo, G. P., Whitaker, J. S., Sardeshmukh, P. D., Giese, B. S., McColl, C., et al. (2019). Towards a more reliable historical reanalysis: Improvements for version 3 of the Twentieth Century Reanalysis system. *Quarterly Journal of the Royal Meteorological Society*, 145(724), 2876–2908. https://doi.org/10.1002/qj.3598
- Svensson, G. (1986). Recognition of peat-forming plant communities from their peat deposits in two south Swedish bog complexes. Vegetatio, 66(2), 95–108. https://doi.org/10.1007/BF00045499
- Taylor, L. S., Swindles, G. T., Morris, P. J., Gałka, M., & Green, S. M. (2019). Evidence for ecosystem state shifts in Alaskan continuous permafrost peatlands in response to recent warming. *Quaternary Science Reviews*, 207, 134–144. https://doi.org/10.1016/j.quascirev.2019. 02.001
- Van der Putten, N., Hébrard, J.-P., Verbruggen, C., Van de Vijver, B., Disnar, J.-R., Spassov, S., et al. (2008). An integrated palaeoenvironmental investigation of a 6200 year old peat sequence from Ile de la Possession, Iles Crozet, sub-Antarctica. *Palaeogeography, Palaeoclimatology, Palaeoecology*, 270(1–2), 179–195. https://doi.org/10.1016/j.palaeo.2008.09.014
- Van der Putten, N., Mauquoy, D., Verbruggen, C., & Björck, S. (2012). Subantarctic peatlands and their potential as palaeoenvironmental and palaeoclimatic archives. *Quaternary International, Peat Stratigraphy and Climate Change*, 268, 65–76. https://doi.org/10.1016/j.quaint.2011. 07.032
- Vorkauf, M., Marty, C., Kahmen, A., & Hiltbrunner, E. (2021). Past and future snowmelt trends in the Swiss Alps: The role of temperature and snowpack. *Climate Change*, *165*(3–4), 44. https://doi.org/10.1007/s10584-021-03027-x
- Wang, Q., Fan, X., & Wang, M. (2014). Recent warming amplification over high elevation regions across the globe. *Climate Dynamics*, 43(1–2), 87–101. https://doi.org/10.1007/s00382-013-1889-3
- Wietrzyk-Pełka, P., Cykowska-Marzencka, B., Maruo, F., Szymański, W., & Węgrzyn, M. (2020). Mosses and liverworts in the glacier forelands and mature tundra of Svalbard (high Arctic): Diversity, ecology, and community composition. *Polish Polar Research*, 41, 151–186. https://doi. org/10.24425/ppr.2020.132571
- Xu, J., Morris, P. J., Liu, J., & Holden, J. (2018). PEATMAP: Refining estimates of global peatland distribution based on a meta-analysis. Catena, 160, 134–140. https://doi.org/10.1016/j.catena.2017.09.010
- Young, D. M., Baird, A. J., Gallego-Sala, A. V., & Loisel, J. (2021). A cautionary tale about using the apparent carbon accumulation rate (aCAR) obtained from peat cores. *Scientific Reports*, 11(1), 9547. https://doi.org/10.1038/s41598-021-88766-8
- Yu, Z., Beilman, D. W., & Jones, M. C. (2009). Sensitivity of northern peatland carbon dynamics to Holocene climate change. Carbon Cycling in Northern Peatlands, 184, 55–69. https://doi.org/10.1029/2008gm000822
- Yu, Z., Beilman, D. W., & Loisel, J. (2016). Transformations of landscape and peat-forming ecosystems in response to late Holocene climate change in the western Antarctic Peninsula. *Geophysical Research Letters*, 43(13), 7186–7195. https://doi.org/10.1002/2016GL069380